Author's response to reviews

Title: A Case Report: Familial Glucocorticoid Deficiency associated with Familial Focal Segmental Glomerulosclerosis.

Authors:

Nanik Ram (nanik.ram@aku.edu)
Ali Asghar (ali.asghar@aku.edu)
Najmul Islam (najmul.islam@aku.edu)

Version: 3 Date: 1 December 2012

Author's response to reviews: see over
Subject: **Re-Submission of case report after 2\textsuperscript{nd} revision.**

Dear Sir/Madam

Following are the corrections have been made according to reviewers report.

**Reviewer: Serap Turan**

1. There is still a question whether steroid deficiency might have role in development of FSGS because both excessive and defective state of cortisol exacerbates the proteinuria in FSGS patients as reported in literature, in fact in our case proteinuria got worsened after steroid replacement.

2. In discussion paragraph starting with “Metwalley KAK et al” has been removed.

Yours Sincerely

Nanik Ram